Author's response to reviews

Title: The spatial distribution of leprosy in four villages in Bangladesh, an observational study

Authors:

Egil E.A.J. Fischer (egil.fischer@wur.nl)
David D. Pahan (chd@dblm.tmbangladesh.org)
Sumanta S.K. Chowdhury (chd@dblm.tmbangladesh.org)
Linda L. Oskam (l.oskam@kit.nl)
Jan Hendrik J.H. Richardus (j.richardus@erasmusmc.nl)

Version: 4 Date: 1 September 2008

Author's response to reviews:

We have removed and checked requested texts